# UC San Diego UC San Diego Previously Published Works

## Title

Gut microbiome diversity, variability, and latent community types compared with shifts in body weight during the freshman year of college in dormitory-housed adolescents.

Permalink

https://escholarship.org/uc/item/0ms3s5ct

**Journal** Gut Microbes, 15(2)

## Authors

Mohr, Alex Ahern, Mary Bruening, Meg <u>et al.</u>

**Publication Date** 

2023-12-01

### DOI

10.1080/19490976.2023.2250482

Peer reviewed

#### **RESEARCH PAPER**



OPEN ACCESS Check for updates

## Gut microbiome diversity, variability, and latent community types compared with shifts in body weight during the freshman year of college in dormitory-housed adolescents

Alex E. Mohr<sup>a,c</sup>, Mary M. Ahern<sup>a</sup>, Dorothy D. Sears<sup>a</sup>, Meg Bruening<sup>a,b</sup>, and Corrie M. Whisner D<sup>a,c</sup>

<sup>a</sup>College of Health Solutions, Arizona State University, Phoenix, AZ, USA; <sup>b</sup>Department of Nutritional Sciences, College of Health and Human Development, Pennsylvania State University, University Park, PA, USA; <sup>c</sup>Center for Health Through Microbiomes, Biodesign Institute, Arizona State University, Tempe, AZ, USA

#### ABSTRACT

Significant human gut microbiome changes during adolescence suggest that microbial community evolution occurs throughout important developmental periods including the transition to college, a typical life phase of weight gain. In this observational longitudinal study of 139 college freshmen living in on-campus dormitories, we tracked changes in the gut microbiome via 16S amplicon sequencing and body weight across a single academic year. Participants were grouped by weight change categories of gain (WG), loss (WL), and maintenance (WM). Upon assessment of the community structure, unweighted and weighted UniFrac metrics revealed significant shifts with substantial variation explained by individual effects within weight change categories. Genera that positively contributed to these associations with weight change included Bacteroides, Blautia, and Bifidobacterium in WG participants and Prevotella and Faecalibacterium in WL and WM participants. Moreover, the Prevotella/Bacteroides ratio was significantly different by weight change category, with WL participants displaying an increased ratio. Importantly, these genera did not display co-dominance nor ease of transition between Prevotella- and Bacteroides-dominated states. We further assessed the overall taxonomic variation, noting the increased stability of the WL compared to the WG microbiome. Finally, we found 30 latent community structures within the microbiome with significant associations with waist circumference, sleep, and dietary factors, with alcohol consumption chief among them. Our findings highlight the high level of individual variation and the importance of initial gut microbiome community structure in college students during a period of major lifestyle changes. Further work is needed to confirm these findings and explore mechanistic relationships between gut microbes and weight change in free-living individuals.

#### PLAIN LANGUAGE SUMMARY

The freshman year of college is a transitional period that may provide insights into the relationship between the gut microbiome and body weight regulation due to the lifestyle changes that increase vulnerability to weight change. During this critical period many of the lifestyle factors that influence body weight formalize and have important bearing on health outcomes throughout an individual's life. In this college-aged population, shifts in community structure and variability of gut microbes were different by weight change trajectory. Genera that underpinned these shifts such as *Bacteroides, Blautia, Bifidobacterium, Prevotella,* and *Faecalibacterium* displayed varying degrees of inter-individual variability and, in some instances, resistance to alternative states. Accounting for these considerations in the context of body weight control in adolescents may prove useful for improving target outcomes in an intervention setting.

#### Introduction

Emerging adulthood is a formative developmental period, hallmarked by independence and substantial changes in living situation and lifestyle behaviors. As a group, incoming college students living in on-campus dormitories typify these factors. Changes in lifestyle behaviors are particularly evident for body weight changes during the first year as unhealthy eating behaviors and low levels of physical activity contribute to approximately 70% of student weight gain during the freshman year

CONTACT Meg Bruening mmb203@psu.edu Department of Nutritional Sciences, College of Health and Human Development, Pennsylvania State University, 110 Chandlee Lab, University Park, PA 16802, USA; Corrie M. Whisner control whisner@asu.edu College of Health Solutions, Arizona State University, 550 N 3rd St, Phoenix, AZ 85004, USA

Supplemental data for this article can be accessed online at https://doi.org/10.1080/19490976.2023.2250482

© 2023 The Author(s). Published with license by Taylor & Francis Group, LLC.

#### **ARTICLE HISTORY**

Received 10 February 2023 Revised 26 June 2023 Accepted 17 August 2023

#### **KEYWORDS**

Microbiota; stability; *Prevotella; Bacteroides;* state transition; Dirichlet allocation; obesity; adolescent; college; diet

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. The terms on which this article has been published allow the posting of the Accepted Manuscript in a repository by the author(s) or with their consent.

(~1–2 kg), often typified as excess weight gain.<sup>1,2</sup> Such changes are problematic for health outcomes as young individuals with overweight and obesity are more likely to remain with overweight or obesity into adulthood.<sup>3</sup>

Differences in the assemblage of gut microbial communities have been broadly observed between individuals with obese and lean body types across different age ranges, including children and adolescents.4,5 These compositional differences remain significantly distinct even when controlling for constitutional factors like genetics.<sup>6</sup> The community landscape of the gut microbiome tracks with a multitude of health features, forming a bidirectional relationship with the host. Notable moderating factors that drive this relationship include diet, physical activity, weight status, and age.<sup>7–10</sup> Therefore, understanding the relationship between the complex community of gut microbes and health over the lifespan is important for improving outcomes, with a shifting focus from cross-sectional to longitudinal investigations. Results from animal models suggest that microbes influence weight gain, but this relationship in humans remains unclear.<sup>11</sup> Indeed, clinical study challenges persist and host-microbiome relationships in humans have been difficult to examine in rigorous investigations.<sup>12,13</sup> For instance, variance at the individual level (e.g., baseline microbiome composition and functional capacity) greatly confounds the interpretation of intervention or environmental exposure outcomes analyses across study populations.<sup>14-16</sup> Differences in host features and gut microbial structure and function are important considerations, particularly when exploring the underpinnings of divergent health trajectories.

In late-stage adolescence as individuals emerge into adulthood, many lifestyle factors that influence excess body weight formalize and have important bearing on health outcomes throughout an individual's life. Age has been identified as an important factor in gut microbiome diversity, with some reports demonstrating statistically significant differences between adolescents and adults.<sup>17,18</sup> Animal studies suggest that adolescents have a malleable microbiome composition that responds more effectively than adults to lifestyle interventions and results in lasting changes in body composition.<sup>19</sup> However, the implications of the relationship between age and community dynamics in the human gut microbiome on the health of adolescents remain relatively unchartered.

Limited longitudinal data are available in which shifts in the gut microbiome in free-living populations can be examined during periods of expected weight gain.<sup>20–22</sup> Such data are particularly sparse in adolescent populations, as their microbiome composition has largely been considered to be similar to that observed in adults.<sup>17,23</sup> Moreover, aggregation of larger, longitudinal sample sets with high dimensionality and variable taxonomic abundances has presented substantial barriers in translating microbial data to meaningful health state insight. Therefore, in this exploratory study, we aimed to evaluate changes in body weight over the freshman year of college in relation to gut microbiome diversity, keystone taxa, and variability. In addition, we sought to explore potential drivers of latent community structure by constructing truncated fractional classification, considering our sizable and variable sample set. The study population was a cohort of incoming freshman college students living in residence halls on campus, grouped for microbiome analysis by weight gain, weight loss, and weight stable classification based on their longitudinal body weight change category. We hypothesized that the assessed microbial markers would distinctly track with each of the three weight change classifications. Findings provide insight into the complex relationship between changes in the gut microbiome community and weight in adolescents.

#### Results

#### Participant characteristics

This study included 139 participants from a larger investigation that has been described previously.<sup>24–26</sup> Briefly, this longitudinal study aimed to understand the impacts of social interactions, friends and friendship networks on nutrition, physical activity, and weight gain during the transition to college life. In the present study, we retained individuals who provided at least two stool samples (sample n = 372) at any time throughout the study duration to support this longitudinal analysis. Baseline participant

Table 1. Baseline demographic and anthropometric characteristics of study participants.

51			<u> </u>	
Characteristic	Total ( <i>n</i> = 139)	WG ( <i>n</i> = 67)	WL ( <i>n</i> = 13)	WM ( <i>n</i> = 59)
Age (years)	18.6 ± 0.7	18.6 ± 0.7	18.5 ± 0.3	18.6 ± 0.6
Sex, % (n)				
Male	30.2 (42)	29.9 (20)	38.5 (5)	28.8 (17)
Female	69.8 (97)	70.1 (47)	61.5 (8)	71.2 (42)
Race/ethnicity % (n)				
White	42.4 (59)	47.8 (32)	23.1 (3)	40.7 (24)
Black	12.9 (18)	13.4 (9)	7.7 (1)	13.6 (8)
Hispanic	25.9 (36)	25.4 (17)	23.1 (3)	27.1 (16)
Other	18.7 (26)	13.4 (9)	46.2 (6)	18.6 (11)
Height (cm)	167.1 ± 10.2	168.9 ± 10.2	168.5 ± 7.8	166.0 ± 10.9
Weight (kg)	68.9 ± 16.9	68.3 ± 16.5	71.8 ± 15.5	69.1 ± 17.9
Waist circumference (cm)	81.5 ± 13.1	80.2 ± 13.3	84.6 ± 12.7	82.2 ± 12.9
Waist/hip ratio	$0.82 \pm 0.07$	$0.81 \pm 0.07$	$0.84 \pm 0.08$	$0.83 \pm 0.07$
BMI (kg/m <sup>2</sup> )	$24.5 \pm 4.9$	$24.1 \pm 5.1$	$25.1 \pm 4.5$	$24.8 \pm 4.8$
BMI Categories (kg/m <sup>2</sup> ), % (n)				
<18.5 kg/m <sup>2</sup>	4.3 (6)	4.5 (3)	7.7 (1)	3.4 (2)
18.5–24.9 kg/m <sup>2</sup>	61.2 (85)	64.2 (43)	46.2 (6)	61.0 (36)
25–29.9 kg/m <sup>2</sup>	23.7 (33)	20.9 (14)	30.8 (4)	25.4 (15)
≥30 kg/m <sup>2</sup>	10.8 (15)	10.4 (7)	15.4 (2)	10.2 (6)
BMI Dichotomized, % (n)				
Not overweight/obese	65.5 (91)	68.7 (46)	53.9 (7)	64.4 (38)
Overweight/obese	34.5 (48)	31.3 (21)	46.1 (6)	35.6 (21)

Data displayed as mean  $\pm$  SD, unless stated otherwise. Abbreviations: WG, weight gain (gain of >3% of baseline weight); WL, weight loss (loss of >3% body weight); WM, weight maintenance (less than or equal to  $\pm$ 3% weight change).

characteristics are described in Table 1 and an overview of the study is displayed in Figure 1a. The median (IQR) number of days between the first and last fecal sample collection was 216 (146.5, 223) days. At baseline, a majority (n = 91, 65.5%) of participants were classified as having a body mass index (BMI) within the normal or underweight category (<25 kg/  $m^{2}$ ), though a substantial number of participants (*n* =48, 34.5%) were classified as overweight or obese  $(BMI \ge 25 \text{ kg/m}^2)$ . Thus, to account for variability in initial body mass, we controlled for baseline BMI. Participants were distributed into three weight change clusters that included weight gain (WG; n =67, 48.2%), weight loss (WL; n = 13, 9.4%), and weight maintenance (WM; n = 59, 42.4%). Specifically, WG, WL, and WM were defined as a gain of >3% of baseline weight, a loss of >3% body weight, and less than or equal to  $\pm 3\%$ , respectively. This classification was based on expert consensus recommending that weight maintenance be considered a weight change of  $\pm 3\%$  of starting body weight, small weight fluctuations between  $\pm 3-5\%$  weight change, and clinically relevant weight change at  $\geq \pm$ 5% change from baseline.<sup>27</sup> Experts recommend that a percentage weight change is preferred over an absolute weight change as it better accounts for baseline body size.<sup>27</sup>

No significant differences between weight change classification at baseline were detected for age, body weight, waist circumference, waist/hip

ratio, or BMI (analysis of variance [ANOVA] tests,  $P \ge .232$ ); nor for self-reported lifestyle behaviors, including moderate to vigorous physical activity (MVPA), screen time, depression, sleep, dietary factors (Kruskal-Wallis tests, and  $P \ge .090$ ; Supplemental Table S1). In addition, the median number of days between the first and last sample collection was not significantly different by weight classification (Kruskal-Wallis test, P =.427). To better understand the relationships between anthropometric, dietary, and lifestyle factors at baseline, we conducted association analyses (total of 91 correlations). Of note, MVPA positively correlated with fruit and vegetable intake (Spearman's rho = 0.466, P.adj = 7.69e-8), while weekly alcohol intake was negatively correlated with sleep (Spearman's rho = -0.221, P.adj =0.039) and positively correlated with added sugar intake (Spearman's rho = 0.219, P.adj = 0.037). Further, red and processed meat intake was positively associated with added sugar intake (Spearman's rho = 0.323, P.adj = 0.006). The mean daily consumption of dietary fiber for males (n =42) and females (n = 97) was  $18.69 \pm 4.93$  g/d and  $14.73 \pm 3.18$  g/d, respectively, which fell below the Adequate Intake for both males (38 g/d) and females (25-26 g/d) per the Dietary Guidelines for Americans.<sup>28</sup>

Change in body weight ( $\Delta$  post- vs. pre-body weight) was significant by weight classification



**Figure 1.** (a) Overview of study design. Participants that experienced weight gain (WG), weight loss (WL), or weight maintenance (WM) over the sampling period were assessed as discrete categories. (b) Change in body weight by weight change category. (c) Pearson's partial correlation analysis between change in body weight and waist circumference. (d) Difference for change in fiber, whole grain (WHGR), fruit and vegetable (F+V), dairy, added sugar, or alcohol intake between WG, WL, and WM participants.

(ANOVA, P = 0.006), with Bonferroni post-hoc comparisons showing significant distinction for all three categories ( $P \le .030$ ; Figure 1b). Participants in the WG category gained an average of 6.6% over their initial body weight (+4.59 ± 2.61 kg: Supplemental Table S2), with the majority gaining over 5% (~70%, n = 45) and eight individuals gaining over 10%. In comparison, the WL group lost 7.1% of their initial body weight (-5.02 ± 2.34 kg), with three individuals losing more than 10% of their baseline body weight. Participants in the WM category maintained a relatively stable body weight, with an overall average weight change of ~0.8% (0.51 ± 1.16 kg). Relatedly, change in waist circumference and waist/hip ratio was significant by group (ANOVA tests, P < .001 & P = .010, respectfully). Bonferroni post-hoc comparisons revealed statistically significant waist circumference differences over the study period for WG participants (+5.44 ± 6.67%) compared to the participants in the WL and WM groups ( $-4.44 \pm 5.06\% \& -0.01 \pm 5.39\%$ , respectively;  $P \le .022$ ). A similar pattern was observed for waist/hip ratio ( $P \le .033$ ) for participants with WG (+2.91 ± 6.03%), when compared to participants with WL ( $-1.04 \pm 5.21\%$ ), and participants with WM (+0.26 ± 5.55%). Partial correlation analysis controlling for baseline BMI, sex, and individual participant revealed a significant positive association between overall change in body weight and waist circumference (Pearson's R = 0.494, P = 9.468e-10; Figure 1c) and, to a lesser degree, overall change in body weight and waist/hip ratio (Pearson's R = 0.203, P = 0.018).

Of note, there were no significant differences between the groups for change in depression (Kruskal-Wallis, P = .656) or moderate-tovigorous physical activity (MVPA; ANOVA, P = .254), with participants maintaining relatively stable self-reported depression scores and physivalues activity relative to baseline cal (Supplemental Table S3). Moreover, there were no significant differences between weight categories for change in nightly sleep hours (Kruskal-Wallis, P = 0.481). Finally, no statistically significant differences were detected for dietary factors over the study period between weight classifications (Kruskal–Wallis tests,  $P \ge .209$ ; Figure 1d; Supplemental Table S3).

# Longitudinal shift in microbial diversity displayed individualization for each weight change trajectory

To evaluate whether weight change over the sampling period was associated with shifts in the gut microbiome, we first calculated observed amplicon sequence variants (ASVs) and Faith's Phylogenetic Diversity (PD) of the 139 participants at their first and last fecal sample collection. Employing linearmixed effect (LME) modeling, observed ASVs and Faith's PD of microbial taxa varied significantly by group (LME,  $P \le 0.050$ ); however, there was no significant difference between the three weight change categories after multiple test correction (P.  $adj \ge 0.087$ ; Figure 2a,b). There were also no significant interaction effects of time and weight classification for either alpha diversity metric (LME,  $P \ge .169$ ). To explore whether participant's initial microbial diversity was associated with shifts in body weight, we constructed multiple regression models regressing change in body weight (post pre-body weight) by baseline alpha diversity, controlling for BMI, sex, and participant. Change in body weight was not significantly associated with the observed ASVs or Faith's PD ( $R^2 \le 0.019, P \ge$ .246), suggesting that initial alpha diversity was not a potential predictive factor for body weight change.

Despite the generally non-significant differences between weight classification groups by alpha diversity, unweighted and weighted UniFrac beta diversity metrics shifted significantly over time (accounting for all time points). Both distances were assessed to account for non-dominant (presence/absence: unweighted UniFac) and dominant (weighted by abundance: weighted UniFrac) microbes and displayed a low correlation to warrant inclusion of both metrics (Mantel test, R = 0.299, P < 2.2e-16; Supplemental Figure S1). Assessing unweighted and weighted UniFrac distances, permutational analysis of variance (PERMANOVA) revealed that most variation in community composition was explained by the individual participants  $(R^2 \ge 0.605, P.adj \le .002;$  Supplemental Tables S4 & S5). Composition also shifted significantly by weight change category over the duration of the study for weighted UniFrac, though this explained much less variation (interaction effect,  $R^2 = 0.011$ , *P.adj* = 0.003) and was not significantly associated with unweighted UniFrac (interaction effect,  $R^2 = 0.005$ , P.adj = .075). In addition, we calculated intraparticipant distances between sequential stool collections and compared these between groups accounting for baseline BMI, sex, and the individual (as a random effect). Both unweighted and weighted UniFrac intra-participant distances decreased after the first pairwise comparison (LME, *P-adj*  $\leq$  1.2e-4) but did not differ by weight change category (*P.adj*  $\geq$ .244; Figure 2c,d). Inter-individual distances were similarly not significant by group (*P.adj*  $\geq$  .549; Supplemental Figure S2A-B), though were greater than intra-individual weighted UniFrac distances  $(+41.31\% \pm 22.53\%, P < 2.2e-16)$  (Figure 2e), confirming that the gut microbiome is individualized as previously reported over a similar timeframe<sup>29</sup>. Indeed, individual effects within weight change categories accounted for greater variation over time for both distance matrices (PERMANOVA,  $R^2 \ge 0.257$ , *P.adj*  $\leq$  .002). At the genus level, microbial shifts across individuals in all three weight change categories were also apparent by visualizing the time change of the first principal coordinates axis (Figure 2f,h; Supplemental Figure S3A-C). We examined the linear coefficients of PERMANOVA models using weighted UniFrac distances to assess



**Figure 2.** Gut microbiome diversity and community structure of freshman college students categorized by weight gain (WG), weight loss (WL), or weight maintenance (WM) change trajectories over the academic year. Alpha diversity, measured by (a) observed ASVs and (b) Faith's PD. (c) Unweighted UniFrac and (d) weighted UniFrac distances within individuals (intra-individual) were calculated for the first ( $\Delta$  first and second samples) and second ( $\Delta$  first and third samples) distances. Median ±95% confidence intervals displayed in black. (e) Boxplots of the intra- and inter-individual weighted UniFrac distances for the 139 participants. Shifts in GM community at the genus level were observed across individuals on the first principal coordinate over the academic year of (f) WG participants, (g) WL participants, and (h) WM participants (weighted UniFrac). Each series of connected points represents a single individual in the study with the number representing the time point of the study in months. Model coefficients of the PERMANOVA analysis (model: Genus ~ weight category × time) of weighted UniFrac distances for the (i) WG participants, (j) WL participants, and (k) WM participants displayed distinction and similarity. Genera with high coefficients (positive or negative) were visualized to distinguish the taxonomic drivers of the microbiome community over time between the three weight change trajectories. Only the top 10 genera are shown. In boxplots, the median is represented by the center line with the box representing the 1st and 3rd quartiles, whiskers extend 1.5× the interquartile range.

taxonomic underpinnings of these shifts by weight change trajectories and dominant influences at the genus level. Sources of variation in WG community composition over time included positive contributions from notable taxa such as *Bacteroides*, *Blautia*, *Bifidobacterium*, and negative contributions from *Prevotella* and *Faecalibacterium* (Figure 2i). In contrast, *Prevotella* and *Faecalibacterium* were positive contributors and *Bifidobacterium* was a negative contributor for the WL and WM microbiomes,

though, as with WG category, *Bacteroides* was dominant (Figures 2j,k).

# The ratio of Prevotella/Bacteroidetes did not significantly change across the WG, WL, and WM microbiomes

Consistent with the genera that contributed to variation in beta diversity over time, stool samples from participants at baseline were dominated by Bacteroides (mean relative abundance: WG, 16.9% ±13.5% SD; WL, 13.9% ±15.7%; WM, 18.2% ± 10.8%), Blautia (WG, 11.9% ± 7.6%; WL, 9.2% ± 6.3%; WM,  $11.8\% \pm 7.5\%$ ), Faecalibacterium (WG, 10.4% ± 9.4%; WL, 12.5% ± 11.6%; WM, 7.3%  $\pm$  6.0%), and *Bifidobacterium* (WG, 9.0%  $\pm$ 9.1%; WL, 7.9% ± 8.6%; WM, 10.3% ± 9.8%; Figure 3a). Notably, Prevotella was much more disparate between groups with participants, with WL displaying a greater abundance (WG,  $2.6\% \pm$ 8.7%; WL, 10.2% ± 17.3%; WM, 3.6% ± 13.1%), consistent with the driving taxa in beta diversity shifts in this weight classification. The broad coverage of Bacteroides and sparse (though pronounced, where occupant) abundance of Prevotella was apparent and selected for further analysis due to relevance of these two genera in weight change<sup>30</sup> and long-term diet.<sup>31</sup> Of the participant's first stool collection, Bacteroides displayed a left-skewed abundance pattern and of all genera had the lowest evidence of multimodality via prediction strength index (PSI) (0.23). Conversely, Prevotella had moderate support for bimodality (PSI = 0.71), supporting that this microbe was much less abundant in the gut microbiome across individuals. Calculating the ratio between these taxa (see Materials and Methods), we did not observe significant effects for time or interaction of time and weight change category (LME,  $P \ge .131$ ; Figure 3b). However, there was a significant effect of weight change category (LME, P = 0.033), with multiple comparison testing revealing a nearly significant difference between participants with WL and WM (P.adj = 0.050).

To better understand the stability of these genera and whether they shifted to an alternative state over the sampling period, we performed calculations of multimodality based on bootstrapped potential analyses, as previously described.<sup>32</sup> Referred to as the 'tipping point' (TP), this analysis estimates the intermediate unstable region where such shifts may occur. Participants that intersected the calculated (TP = 0.013)threshold for *Prevotella* and Bacteroides (TP = 0.002) displayed significant increased temporal deviation compared to participants that did not (Wilcoxon rank sum test, P =2.665e-15 and P = 0.003, respectfully; Figure 3c,d). These results are suggestive of an avoidance of temporal stability at the intermediate abundance range (low vs. high) and a preference for either a Bacteroides- or Prevotella-dominated state. This was also apparent by the negative correlation between these two genera accounting for all samples (Spearman's rho = -0.213, P = 3.5e-5).

Adapting a longitudinal and continuous analytical approach inspired by Levy and colleagues,<sup>33</sup> we examined the potential for individuals in the three weight change categories to shift between community types (i.e., Bacteroides- vs Prevotelladominated). Representing potential codominance between Prevotella and Bacteroides, the intermediate space in the middle of each ternary plot was generally empty (Figure 3e). Visually, there was an apparent curvature of individual trajectories that shifted along a Prevotella- or Bacteroidesdominated gut microbiome, with few transitions between these states. Polynomial regression models constructed for participants with WG, WL, and WM all accurately mapped to this curvature, accounting for  $\geq$ 81.8% of the total variance.

# Intra- and inter-participant taxonomic variation between weight change trajectories

To assess the intra- and inter-participant microbial abundance variation, we aggregated ASV reads to the genus level for samples from all time points (n= 372) after filtering out rare taxa and performing a center log-ratio (CLR) transformation on relative abundance values. As described in the Methods section, the resultant 69 features were then passed independently through linear-mixed effect models (LMM) for all samples and then by WG, WL, and WM status with participant as a random variable. As described recently by Olsson and colleagues (2022), intraclass correlation coefficients (ICC)



**Figure 3.** (a) The relative abundance (%) of bacterial genera in first-year college students' gut microbiome samples. Each vertical bar represents a separate stool sample taken at the beginning of the study period. Samples are categorized by weight gain trajectory over the academic year (weight gain [WG], weight loss [WL], and weight maintenance [WM]) and are ordered according to the abundance of *Bacteroides*. Samples from 139 individuals in total: WG = 67, WL = 13, and WM = 59, respectively. The top 20 genera are displayed with the remaining taxa collapsed into the category "other". (b) Log10 ratio of *Prevotella*-to-*Bacteroides* displayed a significant effect of weight change category (linear-mixed effects model, P = .033). In boxplots, the median is represented by the center line with the box representing the 1st and 3rd quartiles, whiskers extend  $1.5 \times$  the interquartile range with outliers individually plotted. Temporal variation of (c) *Prevotella* and (d) *Bacteroides* represented by horizontal lines (CLR abundance range) over the study duration for each participant. Bold lines indicate potential state-shift by intersecting the estimated tipping point (red dashed line). (e) Ternary plots for WG, WL, and WM participants depicting the relationship between *Prevotella* (dominate at the right vertex), *Bacteroides* (dominate at the left vertex), and the other genera (other = 1 - (Prevotella - Bacteroides); dominate at the top vertex). Curvature of transitions within and between *Prevotella*- and *Bacteroides*-dominated regions is denoted by a polynomial regression line (red) which accounted for the majority of explained variance (WG = 81.8%, WL = 88.5%, WM = 89.9%).

were calculated from the intra- and interindividual variances extracted from each LMM.<sup>29</sup> The median ICC of all samples was 0.59 (IQR = 0.24), demonstrating the variation between individuals was greater than the variation within individuals. However, 20 taxa had an ICC  $\leq 0.50$ , indicating greater variation within participants (Figure 4a). Notable features with the lowest ICC included Prevotella, Bifidobacterium, Ruminococcus gnavus, Collinsella, and Barnesiella (ICC  $\leq$  0.381; Supplemental Table S6). While many of these taxa displayed a non-trivial relative abundance (i.e.,  $\geq 1\%$ ), there was no significant correlation between CLR relative abundance and ICC (Spearman's rho = -0.122, P = .316). Comparing the ICC between the three weight change categories, participants with WL had a significantly greater overall value compared to participants with WG (Wilcoxon rank sum test, P=.023; Figure 4b), suggestive of increased taxonomic stability within individuals while displaying greater distinction between individuals. No significant differences were detected comparing participants with WG to WM or WL to WM ( $P \ge .230$ ). Examining the ICC for each genus of participants with WL, notably high values (ICC  $\geq 0.80$ ) were apparent for Akkermansia, Butyricicoccus, Dorea, DTU089, Eubacterium eligens group, Eubacterium hallii group, Holdemanella, Incertae Sedis, Lachnospira, Lachnospiraceae UCG 004, Marvinbryantia, Ruminococcus torques group, and Sellimonas (Figure 4c; Supplemental Table S7). In comparison, participants with WG had lower values (ICC  $\leq$ 0.50) for notable features such as Anaerostipes, Bacteroides, Faecalibacterium, Prevotella, and Ruminococcus gauvreauii group compared to participants with WL and WM indicating greater within-participant variability. The total variance in abundance was not significantly correlated with genus CLR relative abundance (Spearman's rho = 0.124, P = 0.308). Moreover, there was no significant difference in total variance between weight change categories (Wilcoxon rank sum test,  $P \ge$ .330). Of note, the genera that had the greatest variation (i.e., the least stable) were similar between weight trajectories. These included the outliers (+1.5 IQR) *Parasutterella* and *CAG 352* shared across all three weight categories, and an unclassified Ruminococcaceae genus and *Bifidobacterium* for participants with WG and WM (Figure 4d).

# Latent community types associate with weight change trajectories and lifestyle factors

We derived the community-type likelihood of all individual samples at the genus level utilizing latent Dirichlet allocation (LDA), an unsupervised machine learning approach that allows fractional membership based on the probability of assignment across community types (as opposed to exclusive assignment).<sup>34</sup> First, to assess the predictive potential of community type in weight change, we specified 30 latent community types based on model fit for 160 genera (comprised of 1,169 ASVs) (Supplemental Figure S4). Hierarchical cluster analysis revealed six groupings (Figure 5a), with apparent separation for community types dominated by *Bacteroidetes* (Cluster 4; Type 19), (Cluster Faecalibacterium 5; Type 30), Bifidobacterium (Cluster 6; Type 4), and, to a lesser extent, Prevotella (Cluster 2; Type 28) (Figure 5b). Clusters 1 and 3 contained 19 and 7 community types, respectfully, and displayed a greater degree of shared taxa contributions for each type. For example, the top five genera for community type 18 within cluster 1, were Alistipes, Bacteroides, Clostridia UCG 014, Eubacterium siraeu, and Faecalibacterium. Assignment probability for each of these taxa was within the range of 10-20%. Similar community types with more varied taxa contributions included types 2, 8, 10, 11, 14, 15, 17, 20, 21, 22, 25, and 26.

The per-sample-per-community type probabilities for the first microbiome samples were extracted for each weight change classification controlling for sex and baseline BMI using Dirichlet regression models. This statistical approach was indicated as we assessed multiple predictive variables with outcome variables as the 30 compositional community types (i.e., values across the subgroups add up to 1 for each observation). Comparing per-community-featureprobabilities across weight change categories at



**Figure 4.** Variation of the gut microbiome at the genus level. (a) Relationship between the intra-individual and inter-individual variations of genera for all samples. Dots colored by intraclass correlation coefficients (ICC) and sized according to the center log-ratio (CLR) relative abundance. Genera with an ICC  $\leq$  0.50 are labeled by name. (b) Boxplots of the ICC for each genus within weight gain (WG), weight loss (WL), or weight maintenance (WM) participants (Wilcoxon rank sum test, \**P* = .023). (c) Heatmap of the ICC for each genus by weight change trajectory. (d) Total variance (intra-individual + inter-individual variance) of each genus for WG, WL, and WM participants, plotted against the CLR relative abundance. Outliers for each group are labeled by name. For boxplots, the median is represented by the center line with the box representing the 1st and 3rd quartiles, whiskers extend 1.5× the interquartile range. Features with an asterisk are listed at the family as the genus was unclassified.



**Figure 5.** (a) Dendrogram of the six hierarchical clusters of community types from participants' baseline sample. (b) Plot of the six hierarchical clusters of community types from participant's baseline sample. Each cluster is denoted by a different color and the numbered nodes represent the latent community types. Ellipses for clusters 1 and 3 are 95% confidence intervals. The composition of each of the 30 community types are displayed in Supplemental Figure S4. Ward's minimum variance method was used for hierarchical clustering. (c) Boxplots of the assigned probability per microbiota community type are organized and colored per the six clusters at the top. Below a dot plot illustrates the associations derived from Dirichlet regression models between the 30 latent community types and anthropometric, behavioral, and dietary factors. Each circle represents a separate association with the size indicating the significance (log10 (P-values)) and the color the effect size (hue) with its direction (red: positive; blue: negative). Asterisks within a circle denote a significance after adjustment with the Benjamini–Hochberg method, \**P.Adj* < .05, \*\* *P.Adj* < .01, \*\*\* *P.Adj* < .001. Abbreviations: BW, body weight; BMI, body mass index; F+V, fruit and vegetable intake; R+P meat, red and processed meat.

baseline, type 12 (dominated by *Oscillospiraceae UCG-002*), 22 (dominated by *Lachnospiraceae NK4A13*), 28 (dominated by *Prevotella*), and 30 (dominated by *Faecalibacterium*) had a greater

likelihood of occurrence in participants with WL, whereas type 19 (dominated by *Bacteroides*) had a greater likelihood of occurrence in participants with WG and WM (*P.adj*  $\leq$ 0.041; Supplemental

·····/ ·······························						
Factor	Community	Effect Size	P-Value	P.Adj		
Alcohol	Type 19	0.256	1.16e-12	4.54e-10		
Alcohol	Type 24	0.244	8.79e-09	1.71e-06		
Waist Circumference	Type 19	0.602	2.86e-07	3.71e-05		
Alcohol	Type 4	0.291	8.96e-07	8.74e-05		
Alcohol	Type 7	0.213	1.57e-06	1.22e-04		
Alcohol	Type 30	0.178	1.91e-06	1.24e-04		
Sleep	Type 19	-0.159	3.25e-06	1.81e-04		
Waist Circumference	Type 29	0.410	3.29e-04	.016		
Added Sugar	Type 4	0.223	4.76e-04	.019		
Alcohol	Type 15	0.191	4.56e-04	.019		
F+V	Type 15	-0.208	7.85e-04	.028		
Fiber	Type 4	-0.217	.001	.035		

**Table 2.** Significant community-type associations with anthropometric, behavioral, and dietary factors of all participant samples taken after the baseline collection.

Data ordered and displayed by *P.A.dj* value (descending). Effect sizes are beta-coefficients derived from multivariate-adjusted Dirichlet regression models, controlled for sex and baseline BMI. Abbreviations: F +V, fruit and vegetable intake; *P.A.dj*, adjusted *P*-value using the Benjamini-Hochberg procedure.

Figure S5). We next considered an unbiased assessment, disregarding weight trajectory assignment (e.g., WG, WL, and WM), of the samples collected after baseline might demonstrate unappreciated community patterns. Using GM samples from the second and third longitudinal collection, we assessed anthropometric (i.e., change in percent body weight, waist circumference, and BMI), behavioral (MVPA, depression, and sleep), and dietary (i.e., intake of fiber, whole-grains, fruit and vegetables, dairy, added sugar, red and processed meat, and alcohol) factors with multivariate-adjusted Dirichlet regression models, controlled for sex and baseline BMI. After multiple testing correction, 12 significant associations with microbial community type were detected (Figure 5c). Alcohol intake (number of weekly drinks) was the most frequent, statistically significant correlate to community type (i.e., six significant associations) and, to a lesser extent, waist, sleep, and added sugar and fruit and vegetable intake (i.e.,  $\leq 2$  significant associations;  $P.adj \le .027$ ; Table 2). Waist circumference displayed the greatest effect size (betacoefficient), with positive values from community types 19 (dominated by Bacteroides) and 29 (dominated by Blautia). In addition, sleep was negatively correlated with type 19 (P.Adj = 1.81e-04). Notably, community type 4 (dominated by *Bifidobacterium*) was positively associated with alcohol and sugar intake (*P.Adj*  $\leq$ .019) and negatively associated with fiber intake (P.Adj = .035).

#### Discussion

To our knowledge, this is the first longitudinal study to observe a statistically significant

relationship between weight change and the structure of the gut microbiome in dormitory-dwelling emerging adults during the first year of college. The average weight change of participants was +2.6% (+1.8 kg), with almost half of the participants gaining weight over the course of their first year. As a proxy for body composition, weight gain was significantly associated with waist circumference, indicating a likely increase in body fat.<sup>35</sup> Notably, the type of weight change was associated with different structural shifts within the community of gut microbes and displayed distinct variability in select taxa in our adolescent population. However, in contrast to our original hypothesis, we did not observe an overall distinct tracking of gut microbiome diversity, keystone taxa, and variability in latent structure with each of the three weight change classifications.

As reported previously by Olsson et al. (2022), we observed greater overall inter-individual variation in dissimilarity and taxonomic composition in the gut microbiome.<sup>29</sup> In agreement with Olsson and colleagues, we also noted that differences within individuals over time can be quite variable and may pose a significant source of confounding when examining health outcomes. However, in comparison to their older population (50–65 years) living in a stable situation, our participants underwent a disruptive life change transitioning into independent living on a college campus. Core genera detected in their study such as Prevotella and Bifidobacterium displayed a much greater ICC in comparison to our results. This discrepancy may have been due to the environmental pressures and significant

changes in anthropometric outcomes we noted in our sample. Moreover, Prevotella was relatively sparse in our samples which may have skewed the variation we observed in relation to other taxa. Others, such as the most prevalent genus, Bacteroides, displayed a greater degree of stability. In addition, our results align with the high between-person differences for abundance of Akkermansia from both longitudinal and crosssectional research.<sup>29,36</sup> Such inter-person variability in select microbes like Prevotella, Bacteroides, and Akkermansia are important considerations in the context of body weight control and, when accounted for in an intervention setting, may improve the detection of significant outcome changes.<sup>30,37–39</sup>

Interestingly, weight change was not significantly related to microbial richness or phylogenetic diversity (alpha diversity) between weight change categories. Moreover, baseline alpha diversity was not predictive of weight change. Together, these findings demonstrate that alpha diversity did not have a significant influence on body weight change over the course of the study. Few clinical studies have evaluated changes in microbial diversity in relation to weight change over time which makes comparison to other studies difficult, especially in adolescent or youth populations. Recently, a metaanalysis assessing microbiome-related metrics reported increased alpha diversity levels, pooling 47 trials with 1,916 participants.<sup>40</sup> However, included studies investigated deliberate weight loss regimens and procedures (energy-restricted diets, pharmacotherapy, bariatric surgery) in middle-aged populations that are likely not representative of the nutritional and behavioral stress experienced by college students. More relatedly, small cross-sectional studies (n = 53) have observed differences in richness and evenness between children with BMI percentiles between the 15<sup>th</sup> and 94<sup>th</sup> percentiles.<sup>41</sup> Conversely, a large cohort study of 295 school-aged children (KOALA Cohort Study) found no differences in microbial diversity and richness when comparing children of different weight status 6–7 years after birth.<sup>42</sup> Authors of the KOALA Cohort Study suggest that they did not observe a relationship because they were able to control for more confounding factors (e.g., diet, lifestyle factors) than most studies. While we report similar covariate control and findings, the stability of alpha diversity is noteworthy considering our longitudinal assessment during a period of expected weight gain and major lifestyle changes such as the first year of college.

In comparison, the microbial community significantly shifted, showing a high degree of variation between and within individuals. This shift was more pronounced between the first and second collections (first distance), with dissimilarity decreasing comparing intra-individual differences of the first and third collections (second distance). These results perhaps indicate a shift closer to baseline composition, considering the initial environmental changes experienced by the participants. Future research may explore changes in microbial communities with more frequent sample collections as well as continuing throughout the advancement of college (i.e., sophomore year, independent living off campus, etc.). While community shifts were apparent for all three weight change categories, the genera that drove the changes in participants with WG compared to WL and WM microbial structure over the academic year displayed some distinction. Protracting elements of the WG shift included Bacteroides, Blautia, Bifidobacterium, while detractors were Prevotella and Faecalibacterium. In the case of many of these taxa, we observed the opposite for the microbiome of participants with WM and WL, with positive contributions from Prevotella and Faecalibacterium and negative contributions from Bifidobacterium. Regardless of weight change trajectory, we noted high coverage of Bacteroides in samples. In comparison, the occurrence of Prevotella was much less, but when it was present in a sample, its relative abundance was pronounced. These results align with previous reports from larger adult cohorts,<sup>33,43,44</sup> where there was a clear ecological dichotomy in the presence of these genera. Prevotella-dominated microbiomes were rare in our overall sample, though did appear to be associated with weight change trajectory assessing the Prevotella/Bacteroides (P/B) ratio. This ratio has been previously shown to predict weight loss,<sup>30,37</sup> and is associated with a dietary intake rich in carbohydrates, resistant starch, and fibers, whereas a diet high in fats but low in fibers is associated with a low P/B ratio.<sup>45</sup> In the present study, we noted a significant, though modestly greater P/B ratio in the WL vs. WM microbiome. Interestingly, all participants reported a low consumption of fiber, and we did not detect significant differences in fiber, whole grain, or fruit and vegetable intake between participants with WG, WL, and WM. This may have been related to our dietary data collection tools and future work in this population should use more precise assessment methods such as 24-hour dietary recalls collected in multiple passes by nutrition professionals.

Our results reinforce that the P/B ratio can be resistant to change,<sup>30,33,46</sup> particularly considering our participants were free-living and not enrolled in a controlled dietary intervention. Such resistance has been speculated to be due to a barrier greatly limiting the transition between Prevotellaand Bacteroides-dominated gut communities.<sup>33</sup> Assessment of longitudinal shifts in alternative states of Prevotella and Bacteroides revealed an intermediate unstable region where, once crossed, resulted in a significant change in abundance to an alternative, presumably more stable, state. Shifts were much more apparent with the genus Prevotella, although did occur with Bacteroides when change in abundance crossed a calculated threshold (i.e., the tipping point). The bimodality of Prevotella is in agreement with previous reports including longitudinal assessments,<sup>32,47</sup> and our support that both Prevotellaand data Bacteroides-dominated microbial communities shift more along an apparent gradient,<sup>48</sup> and are highly individualized. Moreover, there was a noticeable barrier for transition between compositional states. This aligns with the anticorrelative relationship between Prevotella and Bacteroides in human microbiome and the resistance to simple transition to alternative stable states. Levy and colleagues postulated that this is based on several host factors and microbiome dynamics that create regions impermissible (resistant) to transit through microbiome state-space.<sup>33</sup> Indeed, we noted this same curvilinear trend in each individual microbiome trajectory after mapping polynomial regression models for the three weight change categories. As hypothesized, to transition to either a Prevotella- or Bacteroidesdominated state, an individual's gut microbiome must deplete both genera and pass through a permissive region in state space.<sup>33</sup> This has important implications for engineering the microbiota in accordance with improving host health, including when to make such attempts in an individual's life cycle, how to modify diet, and when/ where targeted antimicrobial agents, fecal microbiota transplantations, probiotics, and other therapeutics may be best leveraged.

Observational studies such as ours and others report that direct transitions from a Bacteroidesto a Prevotella-dominated state (and vice versa) are more infrequent.<sup>33</sup> Unsurprisingly, spurring this transition is of great therapeutic interest in relation to multiple health outcomes as these genera have been identified as important correlates with diet and aging.<sup>44</sup> More recently, a randomized control trial in obese adolescents (14-18 years old) found that fecal microbiota transplantation from lean donors that had a high P/B ratio was better able to engraft in the recipient's gut, which promoted shifts in community composition.<sup>49</sup> Specifically, from Bacteroides to Prevotella dominance, showcasing the ability of Prevotella taxa to outcompete Bacteroides taxa in ecological dominance. Depending on several host and environmental factors, this may be due to higher bacterial growth rates of Prevotella, depending on internal and external conditions.<sup>50</sup> Interestingly, this pattern of unbalanced occurrence has been observed across multiple body sites in humans, including the airways, gut, skin, oral cavity, and vagina.<sup>51</sup> This occurrence pattern is interesting considering that these genera are phylogenetically related (i.e., from the order Bacteroidales), yet display markedly different functionalities. For example, Prevotella has a higher fiber utilizing capacity and total short-chain fatty acid production.<sup>52</sup> In the context of body weight shifts, a microbiome with a greater P/B ratio appears to support weight decrease when supplied with certain nutritional inputs, such as a high fiber/wholegrain diet.<sup>30,53,54</sup> We did not observe changes in nutritional factors in WG, WL, and WM participants that would promote such a change, including increased dietary fiber, whole-grain, and/or fruit and vegetable intake. This potential lack of priming the functional capacity of a high P/B microbiome may have been partially responsible for some of the observed weight change and could have important implications for diet modification in these individuals.

While we conducted several analyses exploring Bacteroides, Prevotella, and other keystone taxa, this approach may lend oversimplification to the immense dimensionality of microbiome data, especially considering the relatively large number of samples and ASVs captured in our analysis. This was our motivation for dimensionality reduction via the formation of latent communities grouped by co-occurring bacterial taxa and, in addition to, hierarchical clustering. Rather than form discrete notion memberships, forwarded а as 'enterotypes',<sup>43</sup> we sought to leverage LDA modeling for its fractional membership. This tool has been implemented with cross-sectional designs and has shown recent promise for interpreting latent structures within the human gut microbiome.<sup>55,56</sup> The more pronounced community types identified in our analysis, by abundance and hierarchical separation, were dominated by several genera we had previously identified in our PERMANOVA models and relative abundance assessment, such as Bacteroides, Prevotella, Blautia, Bifidobacterium, and Faecalibacterium. The community types that are significantly associated with the microbiome of WL participants, including types 12, 22, 28, and 30, suggest these bacterial communities may play a role in promoting weight loss. Considering the vulnerability of our population, deleterious lifestyle factors (i.e., mood disturbance, dietary intake, alcohol consumption, etc.), and within-healthy range starting BMI, this should not be viewed as a necessarily beneficial outcome. Even so, by percent assigned probability, types 28 and 30 were notable features and were all but dominated by Prevotella (with minor contributions from an unclassified genus from the Lachnospiraceae family, Anaerostipes, Coprococcus, and Sutterella) and Faecalibacterium (with minor contributions from an unclassified genus from the Lachnospiraceae family, Blautia, Coprococcus, and Fusicatenibacter), respectfully. While not the intent of this community analysis, the feature importance of these genera in WL participants is notable as they have been considered commensal fixtures within a healthy gut microexample, species within biome. For the

Faecalibacterium genus, such as F. prausnitzii, are known to have anti-inflammatory properties critical to colonocyte health<sup>57</sup> and are significant butyrate producers.<sup>58</sup> Moreover, both Faecalibacterium and Prevotella have been associated with healthier dietary patterns in Western populations (i.e., lower fat, greater intake of whole grains and fiber, increased diet quality, etc.),<sup>54,59,60</sup> with increased abundance of Prevotella linked to weight loss success.<sup>30,37</sup> Conversely, community type 19, almost completely dominated by Bacteroides (with minor contributions from an unclassified genus from the Lachnospiraceae family, Flavonifractor, Ruminococcus torque, and UBA 1819), was more associated with the microbiome of WM and WG participants. Depending on the dietary input, Bacteroides has been shown to be positively associated with weight gain<sup>61</sup> and a determinant of fat loss success.<sup>30</sup> It should be noted that the genera of Faecalibacterium, Bacteroides, and Prevotella all have diverse species, and this important nuance was likely not captured as we were limited in our taxonomic resolution. Indeed, of the 1,000+ ASVs after filtering, we had condensed to 160 genera prior to community modeling. Regardless, the associations found for the three weight loss trajectories are consistent with patterns at the genus level noted in the context of body weight and composition regulation.

As has been reported in previous work employing LDA modeling<sup>55,56</sup>, there is utility in providing a more truncated fractional classification to the highly complex community of microbes in the gut, especially when assessing larger sample sizes. Our approach sought to afford more meaningful insight into the overall structure of the gut microbiome over time. Longitudinal assessments are critical as capturing shifts in health states with tools like LDA may yield diagnostic value and provision for identification of candidate communities and taxa as therapeutic targets. Importantly, LDA learns patterns of co-occurrence of features rather than clustering observations based on distance measures. The derived community types may cluster based on factors explored here, such as environmental exposure and associated factors moderated by behavior (i.e., stress, habitual diet, etc.). Indeed, upon exploration of the associations between community types and anthropometric,

dietary, and behavioral factors, we noted several important trends relating to body weight and composition modulation. For instance, waist circumference had significant correlations with several of the latent community types, while change in body weight and BMI did not. As a proxy for central adiposity, these findings suggest the significance of increased body fat in this region. Community Type 19, dominated by Bacteroides, and 29 dominated by Blautia (with minor contributions from an unclassified genus from the Ruminococcaceae family, Lachnoclostridium, Monoglobus, and Turicibacter). In rodent models, high-fat diets and bile acid influx in the gut have been associated with Blautia. <sup>62</sup> In humans, Blautia has been associated with visceral fat in a large sample of twins from the TwinsUK cohort (n = 3,666).<sup>63,64</sup> Notably, *Blautia* in our analysis was present in many community types (Types: 2, 3, 6, 8, 9, 11, 13, 15, 21, 23, 25, 26, and 30) with its near exclusive assignment in only one community Type (i.e., type 29). Thus, these results provide support for fractional assignment of LDA to identify the pervasiveness and ability of taxa like Blautia to co-operate/habitate with other microbes, whether due to ecological pressures and requirements, mutually beneficial relationships, function, etc. Indeed, the high frequency of co-occurrence across community types is plausible considering Blautia contains many stress-related genes, can utilize many different sugars, and produces organic acids (chiefly acetic acid as well as some long-chain fatty acids) and an array of secondary metabolites.<sup>65</sup> Many of these secondary metabolites, like non-ribosomal peptides, polyketides, lanthipeptides, and bacteriocins, may function to aid nutrient acquisition, chemical communication, and inhibit/promote colonization of other microbes,<sup>65–67</sup> which reflects in its ability to reside in a diverse range of hosts.<sup>68</sup>

Other significant associations were unsurprising considering the exposures that are present in a college setting, such as increased ingestion of alcohol and sleep disturbance. Of the 12 significant associations observed in the current study, half of those between community types and anthropometric, dietary, and behavioral factors were related to alcohol intake. Excessive alcohol use has been suggested to have a profound effect on the gut microbiome by depleting several health associated

taxa, such as Akkermansia, Faecalibacterium, Prevotella, Ruminococcus, and increases in potendeleterious microbes, tially such as Enterobacteriaceae.<sup>69</sup> Of note, community type 4, which was largely driven by Bifidobacterium, was positively associated with alcohol and sugar intake. Several Bifidobacterium strains have been identified for their high capacity for ethanol metabolism (encoding the enzymes alcohol dehydrogenase and acetaldehyde dehydrogenase) and potential as a probiotic agent for alcohol detoxification in humans.<sup>70-72</sup> However, recent work in mice suggests that ethanol does not significantly contribute to changes in the gut microbiota, rather the shortchain fatty acid acetate that enters the gut after being produced by the liver.<sup>73</sup> Regardless, Bifidobacterium has been found to be positively associated with sucrose intake in patients with diabetes.<sup>74</sup> type 2 Various strains of Bifidobacterium have long been known to ferment various sugars, including Glucose, Lactose, Galactose, Mannitol, and Xylose.<sup>75</sup> Moreover, Bifidobacterium possesses a bifid shunt, which could efficiently produce adenosine triphosphate from glucose.<sup>76</sup>

To the best of our knowledge, no longitudinal studies similar to ours in scale and duration have been completed in emerging adult populations. Many existing studies on gut microbiome structure in relation to weight have focused on adult populations. Overall, these studies among adults have demonstrated that community structure and species richness may be associated with obesity or weight gain over time.<sup>77</sup> However, research indicates significant differences between the adult and adolescent microbiome, and it has been suggested that abundance may drive these differences in community composition.<sup>17,78</sup> Previous studies have also reported that weight is negatively associated with microbial diversity, but few studies have looked at this relationship longitudinally, especially in a vulnerable adolescent population at risk for excessive weight gain. These gaps in the current literature highlight two major strengths of this study: the adolescent population and the longitudinal study design. In particular, a free-living adolescent sample during a time of expected weight gain adds real-world relevance for understanding the interplay between weight change and gut

microbiome composition. Additional strengths of this study include the larger sample size and the racial and ethnic diversities of the sample, enhancing the generalizability of the findings. Limitations of this study include unbalanced samples across the weight change groups as a result of a convenience sample, and high attrition rates for longitudinal fecal sample collection. We assessed change in individual BMIs, an important indicator of health trajectories and often shown to be linked to health behaviors. Several of the analyses presented were limited to categorical classifications of BMI change which may have limited the clinical impact of this work when compared to continuous data approaches. Further, we want to acknowledge limitations in assessing BMI, particularly among a diverse population, and emphasize that in 2023, the American Medical Association indicated that BMI should no longer be used alone as a diagnostic tool for obesity.<sup>79,80</sup> Regardless, this study serves as an important first step in uncovering the many underpinning elements of the relationship between the gut microbiome, lifestyle factors, and weight gain in college students. Further studies assessing the impact of the gut microbiome in mediating the relationship between lifestyle factors and weight gain may help us understand the phenomenon of college weight gain. This understanding may better inform prevention efforts and minimize weight gain throughout early adulthood. Future research should focus on more frequent sampling, incorporation of additional relevant sample types (i.e., fecal and plasma metabolomics), and tighter study control. Researchers may also sample both adults and adolescents in future longitudinal studies analyzing changes in the gut microbiome in relation to changes in weight to compare differences by life stage. A focus on energy balance studies may also be beneficial for elucidating the role of microbes in weight change trajectories.

In conclusion, this longitudinal study highlights the importance of individual variability, drivers of shifts in community structure, and the relationship with nutritional and behavioral factors on the gut microbiome of adolescents of three weight change categories in a college setting. While we show many qualities of an adultassociated assemblage of gut microbes, such as apparent gradients between shifts in taxonomic dominance, we also highlight differential variability. In our well-phenotyped cohort, we note that tracking, let alone, predicting weight change by any of the numerous accepted determinants (i.e., diet, microbiome, starting weights, psychological readouts, etc.) is extremely difficult. While we did note some differential shifts in taxa associated with weight change classification, such results will require reinforcement via replication in a distinct, though similar, adolescent cohort. Regardless, this formative period, prior to adulthood, may ultimately offer an opportunity for targeted microbiome inventions. Future work should seek to better understand the potential malleability at this life-stage and whether certain interventions can be more effectively leveraged here to align an individual's weight-trajectory to a healthy state.

#### **Materials and methods**

#### Participants and study design

The devilWASTE study was a sub-study of the Social impact of Physical Activity and nutRition in College (SPARC) study, which sought to analyze relationships between lifestyle factors, weight outcomes, and the social netstudents.<sup>24</sup> works of first-year college DevilWASTE participants were recruited from the SPARC cohort from six dorms across three different Arizona State University (ASU) campuses. Participants in devilSPARC were recruited from one dorm on each of the three The exclusion criteria campuses. for devilWASTE included age less than 18 years, certain gastrointestinal conditions such as malabsorptive disease, history of eating disorders, antibiotic use 2-3 months prior to study visits, and current conditions that affect the microbiome including HIV infection, diabetes, or high blood pressure. Inclusion criteria included living in a residence hall at ASU, English speaking, and participation in the SPARC study. Eligible participants provided written informed consent before enrollment. The devilWASTE study and the parent SPARC study were approved by the Arizona State University Institutional Review Board.

#### Data collection

Recruitment for devilWASTE took place during the academic year starting in August 2015 and data collection continued through May 2016. All participants were made aware of the study at dormitory floor meetings, after which they provided voluntary, written informed consent. Consent was not obtained until study staff ensured that all questions and concerns had been addressed individually with each potential participant. The study protocol, recruitment, and data collection documents were reviewed and approved by the Arizona State University Institutional Review Board on July 30, 2015 (IRB# 1309009596). Participants provided stool samples at up to three of the four SPARC study time points (beginning and end of fall and spring semesters). In total, we collected 507 stool samples from 272 participants. Due to the expected high attrition rate for longitudinal sampling in a shared living environment (e.g., stigma and highstress environment), we lost 133 participants to follow-up leaving an n of 139 for this analysis. Only individuals who provided at least two stool samples (two samples, n = 45; three samples, n = 94) at any time throughout the study duration were retained in the present study to support this longitudinal analysis. At each time point, anthropometrics, physical activity, and diet information were collected.

Anthropometrics were obtained by trained research staff using Seca 869 scales (Seca, USA) for weight, Seca 217 stadiometers (Seca, USA) for height, and flexible, tension spring-loaded Gulick measuring tapes for waist circumference (Creative Health Products, USA). These measurements were completed up to three times to ensure accuracy. The two measures that were within 0.5 kg and 0.5 cm for weight and height, respectively, were averaged for final measurements. Participant BMIs were calculated and reported in kg/m<sup>2</sup>.

Questionnaires were used to evaluate lifestyle behaviors. Physical activity was assessed with the Godin-Shephard Leisure-Time Physical Activity Questionnaire which categorizes and quantifies activity into vigorous, moderate, and light physical activity.<sup>81</sup> Sedentary behavior was assessed with an additional question: "Yesterday, how much time did you spend in front of a screen (excluding time in class and being physically active)?".<sup>82</sup> Participants selected a response from a range of zero to six hours. Selfreported dietary intake was reported using the National Cancer Institute Dietary Screener Questionnaire that assesses consumption frequency of key food items and food groups.<sup>83</sup> This tool does not estimate caloric intake but rather tracks consumption of food group categories such as fruits and vegetables, high-fat and processed foods, added sugar, dietary fiber, and whole-grain intake. Weekly alcohol intake was assessed by the number of drinks consumed weekly by asking: "For each day of the week in the calendar below, indicate the number of alcoholic drinks typically consumed on that day (Only if yes to alcohol is selected)." The dropdown ranged from 1 to 15 drinks for all days of the week.

#### Fecal sample collection and processing

After anthropometrics and assessments were completed, participants were given a stool sample collection kit (Commode Specimen Collection Kit, Fisher Scientific, Anthem, AZ) and a brief demonstration on how to properly collect a sample. This kit was labeled with a devilWASTE-specific participant ID as well as contact information for the study staff and written instructions. Contents of the kit included a collection bowl and bag that were pre-weighed as well as a cooler and ice pack to keep the sample cold until it reached the Healthy Lifestyles Research Center on the Downtown ASU campus. Participants were instructed to call research staff immediately after collection, so that samples could be retrieved within 60 minutes to avoid bacterial growth and changes in microbial communities. Fecal samples were then stored at -80°C until extraction. Samples were processed after defrosting at 4°C. Between 0.150 and 0.250 g of feces was put into 2 mL PowerBead tubes from a MoBio Power Soil DNA Isolation Kit (12888-100, MoBio, Carlsbad, CA).

MoBio Power Soil DNA Isolation Kits (12888–100, MoBio, Carlsbad, CA) were used to extract microbial DNA from fecal samples in the laboratory of the Principal Investigator (CMW). These kits combine a series of salt and ethanol-based solutions as well as filtering, and centrifugation methods to first decrease the amount of fecal matter in the sample and then break the cell membranes of microbial cells to release the DNA. Due to inherent challenges in extracting DNA from fecal matter which is comprised of a complex mixture of lipids, carbohydrates, salts, and cells, a heating step was incorporated at the beginning of the protocol as well as an additional cleaning protocol to ensure no fecal inhibitors remained in the final DNA sample. These additions are briefly outlined below and were recommended by the extraction kit manufacturer (MoBio, Carlsbad, CA).

The heating step involved heating the PowerBead tube with Bead Solution, fecal sample, and Solution C1 in a heat block at 65°C for 10 min. Heating followed gently inverting the sample tube after the addition of Solution C1. The cleaning protocol involved adding 100 µl of extracted DNA to a new 2 ml collection tube along with 50 µl of Bead Solution, 25 µl of Solution C2, and 25 µl of Solution C3. The sample was vortexed briefly after each reagent addition, followed by centrifugation at  $13,000 \times g$  for 2 minutes at room temperature. Supernatant (~160-190 µl) was transferred to a clean 2 ml collection tube. After gently shaking, 400 µl of Solution C4 was added to the collection tube and vortexed to mix. Tube contents were then applied to the spin filter and centrifuged at  $10,000 \times g$  for 1 minute at room temperature. Filter flow through was discarded. In two rounds, 500 µl of Solution C5 was added to the spin filter and centrifuged at  $10,000 \times g$  for 30 seconds at room temperature. Flow through was discarded and the spin filter was centrifuged at maximum speed for 2 minutes at room temperature. Finally, the DNA was eluted by placing the spin filter in a new 2 ml collection tube, adding 50 µl of Solution C6 to the center of the filter membrane, incubating for 1 minute at room temperature, and centrifuging at room temperature for 1 minute at 10,000  $\times$  g, after which the spin filter was discarded.

Once the DNA was isolated, sample quality and concentration were obtained using a QIAxpert highspeed microfluidic UV/VIS spectrophotometer (9002340, QIAGEN, Germantown, MD). A total of 507 longitudinally obtained samples from 272 participants were extracted and sent for sequencing.

#### Fecal microbiome analysis

Samples were sequenced at the ASU Biodesign Institute Genomics Core Lab. At the lab, sequences were quantified using the Quant-iT PicoGreen assay (P7589, Invitrogen, Carlsbad, CA). Sequencing methods began with amplification through triplicate PCR in 96-well plates followed by next-generation sequencing to identify bacterial features. Amplification of the 16S rRNA gene sequence was completed using primers for the conserved V4 region of the bacterial genome. The V4 region was identified using the forward 515F primers and 806 R reverse primers containing Illumina adaptor sequences. Purification and quantification materials used for PCR in the Genomics Core Lab included the QIAquick PCR Purification (28106, Qiagen, Germantown, MD), and KAPA Library Quantification Kits (KK4824, Kapa Biosystems, Wilmington, MA). After PCR completion, sequencing was carried out on the Illumina MiSeq instrument (SY-410-1003, Illumina, Inc., San Diego, CA). All protocols were completed in accordance with best practices established by the Earth Microbiome Project guidelines.<sup>84</sup>

The 16S rRNA sequencing produced 7,804,945 reads with a median of 20,093 per sample. Paired end demultiplexed sequences were added into the Quantitative Insights into Microbial Ecology 2 (QIIME2) software pipeline and denoised using the DADA2 command (trimmed at position 14 and truncated at position 250) to remove lowquality regions and construct a feature table using ASVs.<sup>85</sup> Next, the ASV feature table was passed through the feature-classifier plugin, which was implemented using a naive Bayes machine-learning classifier, pre-trained to discern taxonomy mapped to the latest version of the rRNA database SILVA (138.1; 99% ASVs from sequences).<sup>86</sup> 515F/806 R region of A phylogenetic tree was then constructed using the fragment-insertion plugin with SILVA at a p-sampling depth of the rarefaction threshold to impute high-quality reads and normalize for uneven sequencing depth between samples. A phyloseq (v1.38.0.) object was then created, and downstream analyses and visualizations were performed in R (v4.1.2.). Sequences identified as Archaea and unclassified Bacteria at the Phyla level, as well as mitochondrial and plant DNA, were removed. Samples were rarified for downstream diversity analysis (7,300 sequences/ sample).

For the diversity analysis, we implemented a phylogenetically informed approach. Specifically, alpha diversity was measured using Faith's PD using the *picante* package (v1.8.2.). In addition, the observed ASVs were calculated to survey the overall richness with the *phyloseq* package. Beta diversity was calculated using weighted UniFrac, placing emphasis on dominant taxa, and unweighted UniFrac, to consider dissimilarity merely on membership. These beta diversity metrics were calculated with the *phyloseq* package.

To assess the community type of each participant sample at the genus level, we applied LDA probabilistic models. A low cutoff for taxa filtering was used (i.e., taxa with a total relative abundance of 1e-5 were filtered out) per the nature of this method, as previously described.<sup>55</sup> Samples were then surveyed to estimate the most preferable number of latent community types using the 'FindTopicsNumber' function in the ldatuning package (v1.0.2.). The scoring algorithms "CaoJuan2009"<sup>87</sup> and "Arun2010"<sup>88</sup> were specified as metrics, and the model fit was established using Gibbs' sampling method. The number of ideal community types was determined to be 30, and per-community type-per-taxa and per-sample-percommunity type were extracted after performing the LDA using the 'LDA' function in the topicsmodels package (v0.2.12.). The per-sample-percommunity type probabilities were multiplied by the read count for each sample. Hierarchical cluster analysis was then performed using an agglomerative nesting approach (based on a lower number of community types) on scale normalized data, as previously reported.55 This was conducted with the 'agnes' function in the *cluster* package (v2.1.4.) using Ward's minimum variance method. The optimal number of clusters was determined with the Dunn Index. Visualization was expressed with a dendrogram and cluster plot with the factoextra package (v1.0.7).

#### Statistical analysis

Anthropometric, behavioral, and dietary data from participants were first assessed for normality using QQ-plots and Shapiro–Wilk's test. Differences in baseline values between weight change categories were then analyzed by ANOVA or Kruskal–Wallis tests, depending on normality. Associations between these factors were assessed with Spearman's rank correlation coefficients. Changes in anthropometric, behavioral, and dietary data were calculated for the first and last collection and assessed between weight change categories by ANOVA or Kruskal–Wallis tests with Bonferroni post-hoc comparisons, where appropriate. The association between change in body weight and waist circumference was assessed by Pearson's partial correlation analysis, controlling for baseline BMI, sex, and individual participant.

For the gut microbiome, alpha diversity metrics were assessed for normality (Shapiro-Wilk's tests) and log-transformed. Next, LME models were used to test the effect of time and the interaction of group and time using the *nLME* package (v3.1.153.). Models included baseline BMI and sex as covariates with participant included as a random effect. For any significant interactions, pairwise testing was performed using the *emmeans* package (v1.8.0.). Associations between baseline alpha diversity and body weight were assessed with multiple regression models using the *lmtest* package (v0.9.40.) to explore potential differential diet responses using the gut microbiome as a determinant. These associations were calculated using baseline alpha diversity metrics and change in body weight (post - pre values), accounting for sex, baseline BMI, and the individual. For beta diversity, the mantel test was utilized to assess the correlation between the weighted and unweighted UniFrac distance matrices for all samples via the vegan package (v2.6.2.) with 999 permutations. Next, PERMANOVA tests were constructed for UniFrac distances in the *vegan* package testing the effects of the individual (nested factor), weight change category, all time points, and the interaction between these factors, controlling for sex and baseline BMI (number of permutations = 999). Contributions from dominant genera to the PERMANOVA models of weighted UniFrac distances were determined by passing the resulting PERMANOVA object through the coefficients function found in the base stats package in R. Intra- and inter-individual compositional variability was calculated between all sequential microbiome samples for both weighted and unweighted

UniFrac distances, log transformed, and compared by weight change categories using LME models accounting for baseline BMI, sex, and the individual (as a random effect). To assess the overall differences between intra- and inter-individual variability, all values were pooled by comparison and assessed via a Wilcoxon rank sum test.

To calculate the ratio between the genera Prevotella and Bacteroides, we converted taxonomic data to relative abundance, added a pseudo count of  $5.0 \times 10^{-7}$  to samples where *Prevotella* was not present (as previously described),<sup>37</sup> performed a log10 transformation on both, and then divided Prevotella by Bacteroides. Employing the nLME package a LME model was used to test the effect of time (first and last time points) and the interaction of group and time with baseline BMI and sex as covariates with participant included as a random effect. For any significant interactions, pairwise testing was performed using the emmeans package. The PSI was calculated for baseline CLR transformed relative abundance of Prevotella and Bacteroides using the 'bimodality' function in the microbiome package (v1.18.0.). A TP analysis was calculated on the CLR transformed relative abundance of Prevotella and Bacteroides for all samples using a bootstrapped potential analysis (n bootstrap iterations = 100) with the microbiome package. The abundance deviation of these genera for each participant was summed and compared by participants that crossed the TP and those that did not using Wilcoxon rank sum tests. To assess the trend of state transition between Prevotella and Bacteroides, we performed polynomial regression with the *nLME* package using the difference in their relative abundance and their sum subtracted from one, as previously described.<sup>33</sup>

Variation of taxa at the genus level was assessed for all samples on CLR-transformed relative abundance of features with a detection of  $\geq$ 1.0% for at least 10% of the samples. Intra- and interindividual variance was extracted for each taxon via LMM, with participant acting as a random variable using the function 'lmer' in the package *Lme4* (v1.1.30.). Total variance was calculated by summing the intra- and inter-individual variance and ICC by dividing the inter-individual variation by the total variation of the relative abundance for each genus. Associations between CLR relative abundance and ICC and total variation were performed using Spearman's rank correlation tests. Differences between weight change categories for ICC and total variance were performed using Wilcoxon rank sum tests.

For the LDA data, associations between predictor variables and community types were assessed using Dirichlet regression models with the DirichletReg package (v0.7.1). Models were first constructed for weight change trajectory with baseline samples of the 30 latent community types as the response variables. Next, the second and third longitudinal samples were pooled, and models were formed for 13 predictors as continuous variables (anthropometric, dietary, and behavioral factors) with the 30 community types. All predictor variables were standardized by their standard deviation, as previously described<sup>55</sup>. Model fit was assessed by building in relevant covariates, which ultimately led to each model being adjusted for sex and initial BMI. An alpha value of <0.05 was used to denote statistical significance, and P-value adjustments were performed where appropriate using the Benjamini-Hochberg (BH) procedure. All statistical analyses were performed in the R environment (v4.1.2.).

#### **Acknowledgments**

The authors are grateful for the time participants generously volunteered for this project.

#### **Disclosure statement**

No potential conflict of interest was reported by the author(s).

#### Funding

This study was supported by the NIH Common Fund from the Office of the Director and the Office of Behavioral and Social Sciences Research, grant number [1DP5OD017910] (PI: M. Bruening).

#### ORCID

Corrie M. Whisner in http://orcid.org/0000-0003-3888-6348

#### **Authors' contributions**

A.M., M.A., M.B., and C.W. formulated the research question and designed the analysis. M.A., M.B., and C.W. conducted research and/or provided essential materials. A.M. performed statistical analysis, interpreted results, and drafted the manuscript. M.A. and C.W. contributed to the statistical analysis and interpretation of results. A.M., M.A., D.S., M.B., and C. W. critically reviewed the manuscript. M.B. and C.W. secured funding and supervised the entire project. All authors were given the opportunity to read and contribute to the text and approved the final manuscript.

#### Data availability statement

Due to the vulnerable nature of adolescent and emerging adult populations, researchers did not seek participant consent to share study data. Therefore, participants of this study did not agree for their data to be shared publicly, and supporting data are not available.

#### References

- Gropper SS, Simmons KP, Connell LJ, Ulrich PV. Changes in body weight, composition, and shape: a 4-year study of college students. Appl Physiol Nutr Metab. 2012;37(6):1118–1123. doi:10.1139/h2012-139.
- Racette SB, Deusinger SS, Strube MJ, Highstein GR, Deusinger RH. Changes in weight and health behaviors from freshman through senior year of college. J Nutr Educ Behav. 2008;40(1):39–42. doi:10.1016/j.jneb.2007. 01.001.
- Simmonds M, Llewellyn A, Owen CG, Woolacott N. Predicting adult obesity from childhood obesity: a systematic review and meta-analysis. Obes Rev. 2016;17(2):95–107. doi:10.1111/obr.12334.
- 4. Wei Y, Liang J, Su Y, Wang J, Amakye WK, Pan J, Chu X, Ma B, Song Y, Li Y, et al. The associations of the gut microbiome composition and short-chain fatty acid concentrations with body fat distribution in children. Clin Nutr. 2021;40(5):3379–3390. doi:10. 1016/j.clnu.2020.11.014.
- 5. Ferrer M, Ruiz A, Lanza F, Haange S, Oberbach A, Till H, Bargiela R, Campoy C, Segura MT, Richter M, et al. Microbiota from the distal guts of lean and obese adolescents exhibit partial functional redundancy besides clear differences in community structure. Environ Microbiol. 2013;15(1):211–226. doi:10.1111/j. 1462-2920.2012.02845.x.
- Turnbaugh PJ, Hamady M, Yatsunenko T, Cantarel BL, Duncan A, Ley RE, Sogin ML, Jones WJ, Roe BA, Affourtit JP, et al. A core gut microbiome in obese and lean twins. Nature. 2008;457(7228):480–484. doi:10.1038/nature07540.
- David LA, Maurice CF, Carmody RN, Gootenberg DB, Button JE, Wolfe BE, Ling AV, Devlin AS, Varma Y,

Fischbach MA, et al. Diet rapidly and reproducibly alters the human gut microbiome. Nature. 2013;505 (7484):559–563. doi:10.1038/nature12820.

- Whisner CM, Maldonado J, Dente B, Krajmalnik-Brown R, Bruening M. Diet, physical activity and screen time but not body mass index are associated with the gut microbiome of a diverse cohort of college students living in university housing: a cross-sectional study. BMC Microbiol. 2018;18(1):210. doi:10.1186/s12866-018-1362-x.
- Asnicar F, Berry SE, Valdes AM, Nguyen LH, Piccinno G, Drew DA, Leeming E, Gibson R, Roy CL, Khatib HA, et al. Microbiome connections with host metabolism and habitual diet from 1,098 deeply phenotyped individuals. Nat Med. 2021;27(2):321–332. doi:10.1038/s41591-020-01183-8.
- Mohr AE, Jasbi P, Wyst KBV, Woerden I, van Shi X, Gu H, Whisner CM, Bruening M. Association of food insecurity on gut microbiome and metabolome profiles in a diverse college-based sample. Sci Rep-Uk. 2022;12 (1):14358. doi:10.1038/s41598-022-18515-y.
- Maruvada P, Leone V, Kaplan LM, Chang EB. The human microbiome and obesity: moving beyond associations. Cell Host & Microbe. 2017;22 (5):589–599. doi:10.1016/j.chom.2017.10.005.
- 12. Flores GE, Caporaso JG, Henley JB, Rideout JR, Domogala D, Chase J, Leff JW, Vázquez-Baeza Y, Gonzalez A, Knight R, et al. Temporal variability is a personalized feature of the human microbiome. Genome Biol. 2014;15(12):531. doi:10.1186/s13059-014-0531-y.
- Leong KSW, Jayasinghe TN, Wilson BC, Derraik JGB, Albert BB, Chiavaroli V, Svirskis DM, Beck KL, Conlon CA, Jiang Y, et al. Effects of fecal microbiome transfer in adolescents with obesity. JAMA Netw Open. 2020;3(12):e2030415. doi:10.1001/jamanetworkopen. 2020.30415.
- 14. Fragiadakis GK, Wastyk HC, Robinson JL, Sonnenburg ED, Sonnenburg JL, Gardner CD. Longterm dietary intervention reveals resilience of the gut microbiota despite changes in diet and weight. Am J Clin Nutrition. 2020;111(6):1127–1136. doi:10.1093/ ajcn/ngaa046.
- Tebani A, Gummesson A, Zhong W, Koistinen IS, Lakshmikanth T, Olsson LM, Boulund F, Neiman M, Stenlund H, Hellström C, et al. Integration of molecular profiles in a longitudinal wellness profiling cohort. Nat Commun. 2020;11(1):4487. doi:10.1038/s41467-020-18148-7.
- 16. Sowah SA, Milanese A, Schübel R, Wirbel J, Kartal E, Johnson TS, Hirche F, Grafetstätter M, Nonnenmacher T, Kirsten R, et al. Calorie restriction improves metabolic state independently of gut microbiome composition: a randomized dietary intervention trial. Genome Med. 2022;14(1):30. doi:10.1186/s13073-022-01030-0.

- Agans R, Rigsbee L, Kenche H, Michail S, Khamis HJ, Paliy O. Distal gut microbiota of adolescent children is different from that of adults. FEMS Microbiol Ecol. 2011;77(2):404–412. doi:10.1111/j.1574-6941.2011. 01120.x.
- Chierico FD, Abbatini F, Russo A, Quagliariello A, Reddel S, Capoccia D, Caccamo R, Corradini SG, Nobili V, Peppo FD, et al. Gut microbiota markers in obese adolescent and adult patients: age-dependent differential patterns. Front Microbiol. 2018;9:1210. doi:10. 3389/fmicb.2018.01210.
- Mika A, Treuren WV, González A, Herrera JJ, Knight R, Fleshner M, Cotter PD. Exercise is more effective at altering gut microbial composition and producing stable changes in lean mass in juvenile versus adult male F344 rats. PLoS One. 2015;10(5):e0125889. doi:10.1371/journal.pone.0125889.
- 20. Yee AL, Miller E, Dishaw LJ, Gordon JM, Ji M, Dutra S, Ho TTB, Gilbert JA, Groer M, Whiteson KL. Longitudinal microbiome composition and stability correlate with increased weight and length of very-lowbirth-weight infants. mSystems. 2019;4(1):e00229–18. doi:10.1128/mSystems.00229-18.
- 21. Shen N, Caixàs A, Ahlers M, Patel K, Gao Z, Dutia R, Blaser MJ, Clemente JC, Laferrère B. Longitudinal changes of microbiome composition and microbial metabolomics after surgical weight loss in individuals with obesity. Surg Obes Relat Dis. 2019;15 (8):1367–1373. doi:10.1016/j.soard.2019.05.038.
- 22. Monteleone AM, Troisi J, Fasano A, Grave RD, Marciello F, Serena G, Calugi S, Scala G, Corrivetti G, Cascino G, et al. Multi-omics data integration in anorexia nervosa patients before and after weight regain: a microbiome-metabolomics investigation. Clin Nutr. 2021;40(3):1137–1146. doi:10.1016/j.clnu.2020.07.021.
- 23. Yatsunenko T, Rey FE, Manary MJ, Trehan I, Dominguez-Bello MG, Contreras M, Magris M, Hidalgo G, Baldassano RN, Anokhin AP, et al. Human gut microbiome viewed across age and geography. Nature. 2012;486(7402):222-227. doi:10. 1038/nature11053.
- 24. Bruening M, Ohri-Vachaspati P, Brewis A, Laska M, Todd M, Hruschka D, Schaefer DR, Whisner CM, Dunton G. Longitudinal social networks impacts on weight and weight-related behaviors assessed using mobile-based ecological momentary assessments: study protocols for the SPARC study. BMC Public Health. 2016;16(1):901. doi:10.1186/s12889-016-3536-5.
- 25. Bruening M, van WI, Schaefer DR, Hruschka D, Brewis A, Whisner CM, Dunton GF, Todd M, Ohri-Vachaspati P, Laska MN, et al. Friendship as a social mechanism influencing body mass index (BMI) among emerging adults. PLoS One. 2018;13(12):e0208894. doi:10.1371/journal.pone.0208894.
- 26. van Woerden I, Hruschka D, Brewis A, Schaefer DR, Bruening M, Apolzan JW. First-year college students' weight change is influenced by their randomly assigned

roommates' BMI. BMI PLoS One. 2020;15(11): e0242681. doi:10.1371/journal.pone.0242681.

- Stevens J, Truesdale KP, McClain JE, Cai J. The definition of weight maintenance. Int J Obesity. 2006;30 (3):391–399. doi:10.1038/sj.ijo.0803175.
- U.S. Department of Agriculture and U.S. Department of Health and Human Services. Dietary Guidelines for Americans, 2020-2025. 9th Edition. December 2020. Available at DietaryGuidelines.gov.
- 29. Olsson LM, Boulund F, Nilsson S, Khan MT, Gummesson A, Fagerberg L, Engstrand L, Perkins R, Uhlén M, Bergström G, et al. Dynamics of the normal gut microbiota: a longitudinal one-year population study in Sweden. Cell Host & Microbe. 2022;30 (5):726–739.e3. doi:10.1016/j.chom.2022.03.002.
- 30. Hjorth MF, Blædel T, Bendtsen LQ, Lorenzen JK, Holm JB, Kiilerich P, Roager HM, Kristiansen K, Larsen LH, Astrup A. Prevotella-to-Bacteroides ratio predicts body weight and fat loss success on 24-week diets varying in macronutrient composition and dietary fiber: results from a post-hoc analysis. Int J Obesity. 2019;43(1):149–157. doi:10.1038/s41366-018-0093-2.
- 31. Wu GD, Chen J, Hoffmann C, Bittinger K, Chen Y-Y, Keilbaugh SA, Bewtra M, Knights D, Walters WA, Knight R, et al. Linking long-term dietary patterns with gut microbial enterotypes. Sci. 2011;334 (6052):105–108. doi:10.1126/science.1208344.
- Lahti L, Salojärvi J, Salonen A, Scheffer M, de Vos WM. Tipping elements in the human intestinal ecosystem. Nat Commun. 2014;5(1):4344. doi:10.1038/ ncomms5344.
- 33. Levy R, Magis AT, Earls JC, Manor O, Wilmanski T, Lovejoy J, Gibbons SM, Omenn GS, Hood L, Price ND. Longitudinal analysis reveals transition barriers between dominant ecological states in the gut microbiome. Proc Natl Acad Sci USA. 2020;117 (24):13839–13845. doi:10.1073/pnas.1922498117.
- Sankaran K, Holmes SP. Latent variable modeling for the microbiome. Biostatistics. 2018;20(4):599-614. doi:10.1093/biostatistics/kxy018.
- 35. Flegal KM, Shepherd JA, Looker AC, Graubard BI, Borrud LG, Ogden CL, Harris TB, Everhart JE, Schenker N. Comparisons of percentage body fat, body mass index, waist circumference, and waist-stature ratio in adults. Am J Clin Nutrit. 2008;89 (2):500–508. doi:10.3945/ajcn.2008.26847.
- 36. Falony G, Joossens M, Vieira-Silva S, Wang J, Darzi Y, Faust K, Kurilshikov A, Bonder MJ, Valles-Colomer M, Vandeputte D, et al. Population-level analysis of gut microbiome variation. Science. 2016;352 (6285):560–564. doi:10.1126/science.aad3503.
- 37. Hjorth MF, Christensen L, Larsen TM, Roager HM, Krych L, Kot W, Nielsen DS, Ritz C, Astrup A. Pretreatment Prevotella-to-bacteroides ratio and salivary amylase gene copy number as prognostic markers for dietary weight loss. Am J Clin Nutrition. 2020;111 (5):1079–1086. doi:10.1093/ajcn/nqaa007.

- Jie Z, Yu X, Liu Y, Sun L, Chen P, Ding Q, Gao Y, Zhang X, Yu M, Liu Y, et al. The baseline gut microbiota directs dieting-induced weight loss trajectories. Gastroenterol. 2021;160(6):2029–2042.e16. doi:10. 1053/j.gastro.2021.01.029.
- 39. Dao MC, Everard A, Aron-Wisnewsky J, Sokolovska N, Prifti E, Verger EO, Kayser BD, Levenez F, Chilloux J, Hoyles L, et al. Akkermansia muciniphila and improved metabolic health during a dietary intervention in obesity: relationship with gut microbiome richness and ecology. Gut. 2016;65(3):426. doi:10.1136/gutjnl-2014-308778.
- 40. Koutoukidis DA, Jebb SA, Zimmerman M, Otunla A, Henry JA, Ferrey A, Schofield E, Kinton J, Aveyard P, Marchesi JR. The association of weight loss with changes in the gut microbiota diversity, composition, and intestinal permeability: a systematic review and meta-analysis. Gut Microbes. 2022;14(1):2020068. doi:10.1080/19490976.2021.2020068.
- Bervoets L, Hoorenbeeck KV, Kortleven I, Noten CV, Hens N, Vael C, Goossens H, Desager KN, Vankerckhoven V. Differences in gut microbiota composition between obese and lean children: a cross-sectional study. Gut Pathog. 2013;5(1):10. doi:10.1186/1757-4749-5-10.
- 42. Mbakwa CA, Hermes GDA, Penders J, Savelkoul PHM, Thijs C, Dagnelie PC, Mommers M, Zoetendal EG, Smidt H, Arts ICW. Gut microbiota and body weight in school-aged children: the KOALA birth cohort study. Obesity. 2018;26(11):1767–1776. doi:10.1002/oby. 22320.
- 43. Arumugam M, Raes J, Pelletier E, Paslier DL, Yamada T, Mende DR, Fernandes GR, Tap J, Bruls T, Batto J-M, et al. Enterotypes of the human gut microbiome. Nature. 2011;473(7346):174–180. doi:10. 1038/nature09944.
- 44. Claesson MJ, Jeffery IB, Conde S, Power SE, O'Connor EM, Cusack S, Harris HMB, Coakley M, Lakshminarayanan B, O'Sullivan O, et al. Gut microbiota composition correlates with diet and health in the elderly. Nature. 2012;488(7410):178–184. doi:10.1038/ nature11319.
- Christensen L, Roager HM, Astrup A, Hjorth MF. Microbial enterotypes in personalized nutrition and obesity management. Am J Clin Nutrition. 2018;108 (4):645–651. doi:10.1093/ajcn/nqy175.
- 46. Roager HM, Licht TR, Poulsen SK, Larsen TM, Bahl MI. Microbial enterotypes, inferred by the prevotella-to-bacteroides ratio, remained stable during a 6-month randomized controlled diet intervention with the new Nordic diet. Appl Environ Microb. 2014;80(3):1142–1149. doi:10.1128/AEM. 03549-13.
- 47. Costea PI, Hildebrand F, Arumugam M, Bäckhed F, Blaser MJ, Bushman FD, de Vos WM, Ehrlich SD, Fraser CM, Hattori M, et al. Enterotypes in the landscape of gut microbial community composition. Nature

Microbiol. 2018;3(1):8-16. doi:10.1038/s41564-017-0072-8.

- Knights D, Ward TL, McKinlay CE, Miller H, Gonzalez A, McDonald D, Knight R. Rethinking "enterotypes". Cell Host & Microbe. 2014;16 (4):433-437. doi:10.1016/j.chom.2014.09.013.
- 49. Wilson BC, Vatanen T, Jayasinghe TN, Leong KSW, Derraik JGB, Albert BB, Chiavaroli V, Svirskis DM, Beck KL, Conlon CA, et al. Strain engraftment competition and functional augmentation in a multi-donor fecal microbiota transplantation trial for obesity. Microbiome. 2021;9(1):107. doi:10.1186/s40168-021-01060-7.
- 50. Diener C, Qin S, Zhou Y, Patwardhan S, Tang L, Lovejoy JC, Magis AT, Price ND, Hood L, Gibbons SM, et al. Baseline gut metagenomic functional gene signature associated with variable weight loss responses following a healthy lifestyle intervention in humans. mSystems. 2021;6(5):e00964–21. doi:10.1128/ mSystems.00964-21.
- 51. Faust K, Sathirapongsasuti JF, Izard J, Segata N, Gevers D, Raes J, Huttenhower C, Ouzounis CA. Microbial co-occurrence relationships in the human microbiome. PLoS Comput Biol. 2012;8(7):e1002606. doi:10.1371/journal.pcbi.1002606.
- 52. Chen T, Long W, Zhang C, Liu S, Zhao L, Hamaker BR. Fiber-utilizing capacity varies in prevotella- versus bacteroides-dominated gut microbiota. Sci Rep-Uk. 2017;7(1):2594. doi:10.1038/s41598-017-02995-4.
- 53. Hjorth MF, Roager HM, Larsen TM, Poulsen SK, Licht TR, Bahl MI, Zohar Y, Astrup A. Pre-treatment microbial prevotella-to-Bacteroides ratio, determines body fat loss success during a 6-month randomized controlled diet intervention. Int J Obesity. 2018;42 (3):580–583. doi:10.1038/ijo.2017.220.
- 54. Kovatcheva-Datchary P, Nilsson A, Akrami R, Lee YS, De Vadder F, Arora T, Hallen A, Martens E, Björck I, Bäckhed F. Dietary fiber-induced improvement in glucose metabolism is associated with increased abundance of Prevotella. Cell Metab. 2015;22(6):971–982. doi:10.1016/j.cmet.2015.10.001.
- 55. Breuninger TA, Wawro N, Breuninger J, Reitmeier S, Clavel T, Six-Merker J, Pestoni G, Rohrmann S, Rathmann W, Peters A, et al. Associations between habitual diet, metabolic disease, and the gut microbiota using latent Dirichlet allocation. Microbiome. 2021;9 (1):61. doi:10.1186/s40168-020-00969-9.
- 56. Hosoda S, Nishijima S, Fukunaga T, Hattori M, Hamada M. Revealing the microbial assemblage structure in the human gut microbiome using latent Dirichlet allocation. Microbiome. 2020;8(1):95. doi:10. 1186/s40168-020-00864-3.
- 57. Sokol H, Pigneur B, Watterlot L, Lakhdari O, Bermúdez-Humarán LG, Gratadoux J-J, Blugeon S, Bridonneau C, Furet J-P, Corthier G, et al. Faecalibacterium prausnitzii is an anti-inflammatory commensal bacterium identified by gut microbiota

analysis of Crohn disease patients. Proc Natl Acad Sci. 2008;105(43):16731-16736. doi:10.1073/pnas.08048 12105.

- Rios-Covian D, Gueimonde M, Duncan SH, Flint HJ, de Los Reyes-Gavilan CG, Cox M. Enhanced butyrate formation by cross-feeding between Faecalibacterium prausnitzii and bifidobacterium adolescentis. FEMS Microbiology Letters. 2015;362(21):fnv176. doi:10. 1093/femsle/fnv176.
- 59. Haro C, Montes-Borrego M, Rangel-Zúñiga OA, Alcalá-Díaz JF, Gómez-Delgado F, Pérez-Martínez P, Delgado-Lista J, Quintana-Navarro GM, Tinahones FJ, Landa BB, et al. Two healthy diets modulate gut microbial community improving insulin sensitivity in a human obese population. J Clin Endocrinol Metabolism. 2016;101 (1):233–242. doi:10.1210/jc.2015-3351.
- 60. Meslier V, Laiola M, Roager HM, Filippis FD, Roume H, Quinquis B, Giacco R, Mennella I, Ferracane R, Pons N, et al. Mediterranean diet intervention in overweight and obese subjects lowers plasma cholesterol and causes changes in the gut microbiome and metabolome independently of energy intake. Gut. 2020;69(7):1258–1268. doi:10.1136/gutjnl-2019-320438.
- 61. Christensen L, Sørensen CV, Wøhlk FU, Kjølbæk L, Astrup A, Sanz Y, Hjorth MF, Benítez-Páez A. Microbial enterotypes beyond genus level: bacteroides species as a predictive biomarker for weight change upon controlled intervention with arabinoxylan oligosaccharides in overweight subjects. Gut Microbes. 2020;12(1):1–16. doi:10.1080/19490976.2020.1847627.
- Lin H, An Y, Tang H, Wang Y. Alterations of bile acids and gut microbiota in obesity induced by high fat diet in rat model. J Agr Food Chem. 2019;67 (13):3624–3632. doi:10.1021/acs.jafc.9b00249.
- 63. Beaumont M, Goodrich JK, Jackson MA, Yet I, Davenport ER, Vieira-Silva S, Debelius J, Pallister T, Mangino M, Raes J, et al. Heritable components of the human fecal microbiome are associated with visceral fat. Genome Biol. 2016;17(1):189. doi:10.1186/s13059-016-1052-7.
- Roy CIL, Beaumont M, Jackson MA, Steves CJ, Spector TD, Bell JT. Heritable components of the human fecal microbiome are associated with visceral fat. Gut Microbes. 2017;9(1):61–67. doi:10.1080/ 19490976.2017.1356556.
- 65. Liu X, Mao B, Gu J, Wu J, Cui S, Wang G, Zhao J, Zhang H, Chen W. Blautia—a new functional genus with potential probiotic properties? Gut Microbes. 2021;13(1):1–21. doi:10.1080/19490976.2021.1875796.
- 66. Kim SG, Becattini S, Moody TU, Shliaha PV, Littmann ER, Seok R, Gjonbalaj M, Eaton V, Fontana E, Amoretti L, et al. Microbiota-derived lantibiotic restores resistance against vancomycin-resistant enterococcus. Nature. 2019;572(7771):665–669. doi:10. 1038/s41586-019-1501-z.
- 67. Caballero S, Kim S, Carter RA, Leiner IM, Sušac B, Miller L, Kim GJ, Ling L, Pamer EG. Cooperating

commensals restore colonization resistance to vancomycin-resistant enterococcus faecium. Cell Host & Microbe. 2017;21(5):592–602.e4. doi:10.1016/j.chom. 2017.04.002.

- Eren AM, Sogin ML, Morrison HG, Vineis JH, Fisher JC, Newton RJ, McLellan SL. A single genus in the gut microbiome reflects host preference and specificity. ISME J. 2015;9(1):90–100. doi:10.1038/ ismej.2014.97.
- 69. Litwinowicz K, Choroszy M, Waszczuk E. Changes in the composition of the human intestinal microbiome in alcohol use disorder: a systematic review. Am J Drug Alcohol Abus. 2019;46(1):4–12. doi:10.1080/00952990. 2019.1669629.
- Nosova T, Jousimies-Somer H, Jokelainen K, Heine R, Salaspuro M. Acetaldehyde production and metabolism by human indigenous and probiotic lactobacillus and bifidobacterium strains. Alcohol And Alcoholism. 2000;35 (6):561–568. doi:10.1093/alcalc/35.6.561.
- Lim T-J, Lim S, Yoon JH, Chung MJ. Effects of multi-species probiotic supplementation on alcohol metabolism in rats. J Microbiol. 2021;59(4):417–425. doi:10.1007/s12275-021-0573-2.
- 72. Jung S-J, Hwang J-H, Park E-O, Lee S-O, Chung Y-J, Chung M-J, Lim S, Lim T-J, Ha Y, Park B-H, et al. Regulation of alcohol and acetaldehyde metabolism by a mixture of lactobacillus and bifidobacterium species in human. Nutrients. 2021;13(6):1875. doi:10.3390/ nu13061875.
- 73. Martino C, Zaramela LS, Gao B, Embree M, Tarasova J, Parker SJ, Wang Y, Chu H, Chen P, Lee K-C, et al. Acetate reprograms gut microbiota during alcohol consumption. Nat Commun. 2022;13(1):4630. doi:10. 1038/s41467-022-31973-2.
- 74. Hashimoto Y, Hamaguchi M, Kaji A, Sakai R, Osaka T, Inoue R, Kashiwagi S, Mizushima K, Uchiyama K, Takagi T, et al. Intake of sucrose affects gut dysbiosis in patients with type 2 diabetes. J Diabetes Invest. 2020;11(6):1623-1634. doi:10.1111/jdi.13293.
- 75. de VW, Stouthamer AH. Fermentation of glucose, lactose, galactose, mannitol, and xylose by bifidobacteria. J Bacteriol. 1968;96(2):472–478. doi:10.1128/jb.96.2. 472-478.1968.
- 76. Suzuki R, Katayama T, Kim B-J, Wakagi T, Shoun H, Ashida H, Yamamoto K, Fushinobu S. Crystal structures of phosphoketolase thiamine diphosphate-dependent dehydration mechanism\*. J Biol Chem. 2010;285 (44):34279–34287. doi:10.1074/jbc.M110.156281.
- 77. Chatelier EL, Nielsen T, Qin J, Prifti E, Hildebrand F, Falony G, Almeida M, Arumugam M, Batto J-M, Kennedy S, et al. Richness of human gut microbiome correlates with metabolic markers. Nature. 2013;500 (7464):541–546. doi:10.1038/nature12506.
- Hollister EB, Riehle K, Luna RA, Weidler EM, Rubio-Gonzales M, Mistretta T-A, Raza S, Doddapaneni HV, Metcalf GA, Muzny DM, et al. Structure and function

of the healthy pre-adolescent pediatric gut microbiome. Microbiome. 2015;3(1):36. doi:10.1186/s40168-015-0101-x.

- 79. Humphreys S. The unethical use of BMI in contemporary general practice. Br J Gen Pr. 2010;60(578):696–697. doi:10.3399/bjgp10X515548.
- Berg S. AMA: Use of BMI alone is an imperfect clinical measure [Internet]. American Medical Association; 2023 [accessed 2023 Jun 20]. https://www.ama-assn. org/delivering-care/public-health/ama-use-bmi-aloneimperfect-clinical-measure.
- Sirard JR, Pate RR. Physical activity assessment in children and adolescents. Sports Med. 2001;31(6):439–454. doi:10.2165/00007256-200131060-00004.
- Utter J, Neumark-Sztainer D, Jeffery R, Story M. Couch potatoes or French fries: are sedentary behaviors associated with body mass index, physical activity, and dietary behaviors among adolescents? J Am Diet Assoc. 2003;103 (10):1298–1305. doi:10.1016/S0002-8223(03)01079-4.
- National Cancer Institute. Dietary Screener Questionnaire in the NHANES 2009–10 [Internet]; 2009 [accessed 2022 Jun 15]. http://appliedresearch.can cer.gov/nhanes/dietscreen/dsq\_english.pdf.
- 84. Thompson LR, Sanders JG, McDonald D, Amir A, Ladau J, Locey KJ, Prill RJ, Tripathi A, Gibbons SM,

Ackermann G, et al. A communal catalogue reveals earth's multiscale microbial diversity. Nature. 2017;551(7681):457–463. doi:10.1038/nature24621.

- 85. Bolyen E, Rideout JR, Dillon MR, Bokulich NA, Abnet CC, Al-Ghalith GA, Alexander H, Alm EJ, Arumugam M, Asnicar F, et al. Reproducible, interactive, scalable and extensible microbiome data science using QIIME 2. Nat Biotechnol. 2019;37 (8):852–857. doi:10.1038/s41587-019-0209-9.
- 86. Quast C, Pruesse E, Yilmaz P, Gerken J, Schweer T, Yarza P, Peplies J, Glöckner FO. The SILVA ribosomal RNA gene database project: improved data processing and web-based tools. Nucleic Acids Res. 2012;41(D1): D590–6. doi:10.1093/nar/gks1219.
- Cao J, Xia T, Li J, Zhang Y, Tang S. A density-based method for adaptive LDA model selection. Neurocomputing. 2009;72(7–9):1775–1781. doi:10. 1016/j.neucom.2008.06.011.
- Arun R, Suresh V, Madhavan CEV, Murthy MNN. Advances in Knowledge Discovery and Data Mining, Part I, 14th Pacific-Asia Conference, PAKDD 2010, Hyderabat, India, June 21-24, 2010, Proceedings. In: Zaki MJ, Yu JX, Ravindran B, Pudi V, editors. Berlin Heidelberg: Springer; 2010. p. 391–402. doi:10.1007/ 978-3-642-13657-3.